Author's response to reviews

Title: Sources, perceived usefulness and understanding of information disseminated to families who entered home quarantine during the H1N1 pandemic in Victoria, Australia: a cross-sectional study

Authors:

Anne M Kavanagh (a.kavanagh@unimelb.edu.au)
Rebecca J Bentley (bri@unimelb.edu.au)
Kate E Mason (kmason@unimelb.edu.au)
Jodie McVernon (j.mcvernon@unimelb.edu.au)
Sylvia Petrony (spetrony@unimelb.edu.au)
James Fielding (James.Fielding@mh.org.au)
Anthony D LaMontagne (alamonta@unimelb.edu.au)
David M Studdert (d.studdert@unimelb.edu.au)

Version: 4 Date: 20 December 2010

Author's response to reviews: see over
17 December 2010

Dear Editor,

We thank you for accepting for publication our manuscript “Sources, perceived usefulness and understanding of information disseminated to families who entered home quarantine during the H1N1 pandemic in Victoria, Australia: A cross-sectional study”.

Please find below the details of the requested revisions we have made in the proofing stage. We have also checked the manuscript for typographical errors and, accordingly, made some minor revisions.

Yours sincerely,

Anne Kavanagh

**Responses to editor’s requests for changes to final manuscript**

*Author list:* the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.

**Response:** Authors are now listed identically in submission system and the manuscript

*References:* References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].'

**Response:** The correct referencing style is used in the final manuscript, and all references come before the full stop.
Affiliations: Please remember to include the city in the affiliation details.

Response: Cities are now included in the author affiliations listed on the title page.

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Response: All highlighting and tracking has been removed from final manuscript.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Response: Figure 1 has been cropped and uploaded as a PPT file.